Author's response to reviews

Title: A 19-year-old woman presenting a glioblastoma in the cerebellum: a case report.

Authors:

Moulay Rachid m.r.El M El MAAQILI Dr (rachidelmaaqili@hotmail.com)
Adil a.H HOSSINI Dr (adilhossini@gmail.com)
Nizar n.El F EL FATIMI Dr (nizarus@hotmail.com)
Rachid r.G GANA Dr (rachidgana96@hotmail.com)
Said s.A AMAR Dr (saidianapath@yahoo.fr)
Mohammed m.J JIDDANE Dr (Jiddane@hotmail.com)
Fouad f.B BELLAKHDAR Dr (f.bellakhdar@menara.ma)

Version: 4 Date: 26 May 2012

Author's response to reviews: see over
May, 25th, 2012

Dr HOSSINI Adil

Department of Neurosurgery,

Avicenne University Hospital, Rabat, Morocco

e-mail: adilhossini401@gmail.com

Dear Sir,

Please find below all the corrections made for my manuscript. A native English speaking colleague helped me to made the language corrections.

Sincerely

Adil hossini, MD